Author's response to reviews

Title: Alterations in PTEN and PIK3CA in colorectal cancers in the EPIC Norfolk study: associations with clinicopathological and dietary factors

Authors:

Adam Naguib (anaguib@cshl.edu)
James C Cooke (jcc@mrc-mbu.cam.ac.uk)
Lisa Happerfield (lh224@cam.ac.uk)
Lucy Kerr (lucy.kerr@addenbrookes.nhs.uk)
Laura J Gay (lj.gay@qmul.ac.uk)
Robert N Luben (robert.luben@phpc.cam.ac.uk)
Richard Y Ball (richard.ball@nnuh.nhs.uk)
Panagiota N Mitrou (g.mitrou@wcrf.org)
Alison McTaggart (alison@srl.cam.ac.uk)
Mark J Arends (mja40@cam.ac.uk)

Version: 4 Date: 16 March 2011

Author's response to reviews:

Proofing of the manuscript has been completed. A small amount of typographical errors were remedied and two observed spelling mistakes.